Author's response to reviews

Title: Patients with Rheumatoid Arthritis have an altered circulatory Aggrecan Profile

Authors:

Jean C Rousseau (rosseaue@lyon.inserm.fr)
Eren U Sumer (eus@nordicbioscience.com)
Gert Hein (Gert.Hein@med.uni-jena.de)
Bodil C Sondergaard (bcs@nordicbioscience.com)
Suzi Madsen (shm@nordicbioscience.com)
Christian Pedersen (CP@nordjylland amt.dk)
Thomas Neumann (Thomas.Neumann@med.uni-jena.de)
Andreas Mueller (Thomas.Neumann@med.uni-jena.de)
Per Qvist (pq@nordicbioscience.com)
Pierre Delmas (delmas@lyon.inserm.fr)
Morten A Karsdal (mk@nordicbioscience.com)

Version: 4 Date: 7 May 2008

Author's response to reviews:

The authors are written identically and in the order in which they are listed in the submission system.

The qualifications of the authors are removed from the manuscript and submissions system.

The email addresses are on a new line and preceded by the author's initials.

Keywords and Running title are removed.

Section headings and subheadings are changed (fonts, sizes and capitals) to represent the template from BMC.

“Introduction” is changed to “Background” in both the abstract and manuscript.

“Materials & Methods” is changed to "Methods" in both the abstract and manuscript.

Abbreviations are presented as a paragraph instead of a list.

“Authors contributions” is changed to “Authors' contributions”.

All the blank pages are removed.

Figure 5 has been cropped.